Author’s response to reviews

Title: Atrial septal defect (ASD) in a patient with congenital disorder of glycosylation type 1a (CDG-1a): a case report

Authors:

Wu Ruohao (708933019@qq.com)
Li Dongfang (ldf201310@163.com)
Tang Wenting (Tangwt@sysucc.org.cn)
Qiu Kunyin (398067206@qq.com)
Li Yu (761475697@qq.com)
Liao Xiongyu (534980497@qq.com)
Tang Danxia (skyeagle110@qq.com)
Qin Lijun (13600031960@163.com)
Deng Bingqing (bingqingdeng@sina.com)
Luo Xiangyang (luoxy33@126.com)

Version: 2 Date: 06 Nov 2017

Author’s response to reviews:

EDITOR’S COMMENTS:

Q:I have some concern regarding the revision. The question of ASD caused by this defect is a little farfetched and was spelled to them. They agreed to it but their conclusion still relates it casually.

Hence they should highlight the same, possibly change their title.

R: According to the editor's comments, we change our manuscript's title to "Atrial septal defect (ASD) in a patient with congenital disorder of glycosylation type 1a (CDG-1a): a case report" accordingly.